Author’s response to reviews

Title: Low serum IGF1 is associated with hypertension and predicts early cardiovascular events in women with rheumatoid arthritis

Authors:
Malin Erlandsson (malin.erlandsson@rheuma.gu.se)
Lovisa Lyngfelt (lovisa.lyngfelt@gu.se)
N Åberg (david.aberg@medic.gu.se)
Caroline Wasén (caroline.wasen@rheuma.gu.se)
Rachelle Espino (w4m29@students.keele.ac.uk)
Sofia Silfverswärd (sofia.silfversward@rheuma.gu.se)
Mitra Nadali (mitra.nadali@vregion.se)
Katharina Jood (katarina.jood@neuro.gu.se)
Karin Andersson (karin.andersson@rheuma.gu.se)
Rille Pullerits (rille.pullerits@rheuma.gu.se)
Maria Bokarewa (maria.bokarewa@rheuma.gu.se)

Version: 1 Date: 12 Jun 2019

Author’s response to reviews:

Dear Editor

Thanks for letting us improve the manuscript. We have now corrected the headings according to your instructions, added a list of abbreviations and added in the requested parts in the list of declarations. The figures are now in eps format and we have reminded the authors in question to confirm the authorship, I hope you will receive the response very soon. Changes in the manuscript text are marked with blue text.

We now believe the manuscript is ready for publication, if something still needs improvement, please contact me and I will sort it out =)

Best regards
Malin Erlandsson